Author's response to reviews

Title: Developing theory-informed behaviour change interventions to implement evidence into practice: a systematic approach using the Theoretical Domains Framework

Authors:

Simon D French (s.french@unimelb.edu.au)
Sally E Green (sally.green@monash.edu)
Denise A O'Connor (denise.oconnor@monash.edu)
Joanne E McKenzie (joanne.mckenzie@monash.edu)
Jill J Francis (j.francis@abdn.ac.uk)
Susan Michie (s.michie@ucl.ac.uk)
Rachelle Buchbinder (rachelle.buchbinder@monash.edu)
Peter Schattner (peter.schattner@monash.edu)
Neil Spike (neil.spike@vma.com.au)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)

Version: 3 Date: 20 December 2011

Author's response to reviews:

MS: 3491648315818041
Developing theory-informed behaviour change interventions to implement evidence into practice: a systematic approach using the Theoretical Domains Framework

Authors: Simon D French, Sally E Green, et al.

Dear Martin,

Thank you for your editorial comments. We have addressed each of these as follows:

1. In the Background and Methods sections we have addressed the tense and “instructional” sense of writing by more clearly stating what we did.

2. In the Results section we have described the results of what we did.

3. Any suggestions and recommendations for how others should consider this method have been placed in the Discussion.

4. We updated the Abstract to reflect the above changes to the manuscript

5. We have deleted reference to the Series' introductory article

6. We have deleted the Authors' information section

7. We have identified Jeremy Grimshaw as an editorial board member in the competing interests section

One question – we have kept the manuscript type as “Methodology”, but think
perhaps it is now better suited as a “Research” manuscript? Let me know if you want me to change this.

As you suggested we have uploaded a “clean” version of the manuscript. I trust we have now addressed all your comments and suggestions adequately. Let me know if further changes are required.

Kind regards
Simon French, on behalf of the IMPLEMENT author team